Author's response to reviews

Title: Relationship of moderate and low isometric lumbar extension through architectural and muscular activity variables: a cross sectional study

Authors:

Antonio I Cuesta-Vargas (acuesta@uma.es)
Manuel Gonzalez-Sanchez (mgsa@yahoo.it)

Version: 2 Date: 26 April 2013

Author's response to reviews: see over
Dear colleague

I enclose the original of the paper entitled “Relationship of moderate and low isometric lumbar extension through architectural and muscular activity variables: a cross sectional study.” to be consider for publication in the BMC Musculoskeletal Disorders.

The Corresponding Author had full access to all the data in the study and takes responsibility for the integrity of the data and the accuracy of the data analysis as well as the decision to submit for publication.

Authors declare that all authors were fully involved in the study and preparation of the manuscript submitted to the BMC Medical Imaging and that the material within has not been and will not be submitted for publication elsewhere.

Conflicts of interest: The authors declare that they haven’t any commercial relationships which may lead to a conflict of interests.

Yours sincerely

Cuesta-Vargas, A.I., Pt, PhD.
School of Clinical Sciences
Faculty of Health
Queensland University of Technology
Australia